Author's response to reviews

Title: Determining the prevalence of palliative needs and exploring screening accuracy of depression and anxiety items of the Integrated Palliative care Outcome Scale – a multi-centre study

Authors:

Bárbara Antunes (bc521@medschl.cam.ac.uk)

Pedro Rodrigues (pprodrigues@med.up.pt)

Irene Higginson (irene.higginson@kcl.ac.uk)

Pedro Ferreira (pedrof@fe.uc.pt)

Version: 2 Date: 28 Apr 2020

Author's response to reviews:

Please see the point by point docx document uploaded